Author's response to reviews

Title: Undergraduate educational environment, perceived preparedness for postgraduate clinical training, and pass rate on the National Medical Licensure Examination in Japan

Authors:

Yasuhiro Tokuda (tokuyasu@orange.ocn.ne.jp)
Eiji Goto (egotoh@xa2.so-net.ne.jp)
Junji Otaki (jo-tky@umin.ac.jp)
Joshua Jacobs (jjacobs@hawaii.edu)
Fumio Omata (f_omata@mac.com)
Haruo Obara (haruo.obara@gmail.com)
Mina Shapiro (minashap@luke.or.jp)
Kumiko Soejima (soekumi@luke.or.jp)
Yasushi Ishida (yaishida@luke.or.jp)
Sachiko Ohde (sohde@luke.or.jp)
Osamu Takahashi (otakahas@luke.or.jp)
Tsuguya Fukui (fkts@luke.or.jp)

Version: 7 Date: 15 May 2010

Author's response to reviews:

We have revised the tables accordingly.